\begin{table}%t1
%\centering
\par
\caption{\label{tbl-1}Parameters derived from Gaussian fits to the $^{13}$CO $J=2{-}1, J=1{-}0$, and
[C~{\sc i}] $J=1{-}0$ emission line profiles observed towards Galactic molecular clouds.}
$$
\begin{tabular}{l c r@{.}l r@{.}l r@{.}l r@{.}l r@{.}l l}
\hline
\hline
\noalign{\smallskip}
\multicolumn{1}{c}{Source} & No. &
\multicolumn{2}{c}{$V_{\rm LSR}^{\rm CO}$,} &
\multicolumn{2}{c}{$\sigma^{\rm CO}_v$,} &
\multicolumn{2}{c}{$V^{\rm C~I}_{\rm LSR}$,} &
\multicolumn{2}{c}{$\sigma^{\rm C~I}_v$,} &
\multicolumn{2}{c}{$\Delta V,$} & \multicolumn{1}{c}{Ref.} \\
 & & \multicolumn{2}{c}{\kms} & \multicolumn{2}{c}{\kms} &
\multicolumn{2}{c}{\kms}&\multicolumn{2}{c}{\kms}
&\multicolumn{2}{c}{\kms}&\\[-2pt]
\multicolumn{1}{c}{$\scriptscriptstyle (1)$} &
\multicolumn{1}{c}{$\scriptscriptstyle (2)$} &
\multicolumn{2}{c}{$\scriptscriptstyle (3)$} &
\multicolumn{2}{c}{$\scriptscriptstyle (4)$} &
\multicolumn{2}{c}{$\scriptscriptstyle (5)$} &
\multicolumn{2}{c}{$\scriptscriptstyle (6)$} &
\multicolumn{2}{c}{$\scriptscriptstyle (7)$} &
\multicolumn{1}{c}{$\scriptscriptstyle (8)$} \\
\noalign{\smallskip}
\hline
\noalign{\medskip}
TMC-1&1 & 6&1(1)\tablefootmark{a}&2&0(1)&6&0(1)&1&5(2)&0&1(1)& Schilke \etal\ (\cite{Schilke1995})\\[-2pt]
     &2 & 6&1(1)\tablefootmark{a} & 1&6(1) & 6&1(1) & 1&5(1) & 0&0(1) & \\[-2pt]
     &3 & 6&1(1)\tablefootmark{a} & 1&7(1) & 5&9(1) & 1&6(1) & 0&2(1) & \\[-2pt]
     &4 & 6&1(1)\tablefootmark{a} & 1&6(1) & 5&8(1) & 1&2(2) & 0&3(1) & \\[-2pt]
     &5 & 6&2(1)\tablefootmark{a} & 1&5(1) & 6&4(1) & 2&0(2) & --0&2(1) & \\
\noalign{\smallskip}
L183 &6  & 2&83(3)\tablefootmark{a} &2&00(7) &2&2(2)&1&7(3) & 0&63(20) & Stark \etal\ (\cite{Stark1996}) \\[-2pt]
     &7  & 2&41(4)\tablefootmark{a} &1&81(9) &1&6(2)&2&2(4) & 0&81(20) & \\[-2pt]
     &8  & 2&16(3)\tablefootmark{a} &1&84(7) &2&2(1)&1&6(3) & --0&04(10) \\[-2pt]
     &9  & 2&05(4)\tablefootmark{a} & 2&6(1) &1&6(1)&1&8(3) & 0&45(11) & \\[-2pt]
     &10 & 2&20(4)\tablefootmark{a} & 2&0(1) &2&4(1)&1&4(3) & --0&20(11) & \\
\noalign{\smallskip}
Ceph B&11&--13&4(1)\tablefootmark{a} & 2&1      &--14&1(1)    &1&9    &0&7(1) & Mookerjea \etal\ (\cite{Mookerjea2006a})\\
\noalign{\smallskip}
Ori A,B&12&8&6(1)\tablefootmark{b}&4&0(1)&9&3(1)&4&7(1)&--0&7(1)& Ikeda \etal\  (\cite{Ikeda2002})\\[-2pt]
       &13&7&1(1)\tablefootmark{b}&2&4(1)&7&6(1)&3&4(1)&--0&5(1)& \\[-2pt]
       &14&4&8(1)\tablefootmark{b}&2&5(1)&5&2(2)&3&3(4)&--0&4(2)& \\[-2pt]
       &15&10&4(1)\tablefootmark{b}&2&9(1)&10&9(3)&3&3(1)&--0&5(3)& \\[-2pt]
       &16&11&2(1)\tablefootmark{b}&2&6(1)&11&0(1)&3&3(1)&0&2(1)& \\[-2pt]
       &17&9&6(1)\tablefootmark{b} &2&7(1)&9&9(1)&3&2(1)&--0&3(1)& \\[-2pt]
      &18&9&6(1)\tablefootmark{b}&4&4(1)&9&5(1)&5&0(1)&0&1(1)& \\[-2pt]
      &19&8&1(1)\tablefootmark{b}&3&3(1)&8&3(1)&3&7(1)&--0&2(1)& \\[-2pt]
      &20&5&3(1)\tablefootmark{b}&2&3(1)&5&5(1)&3&0(1)&--0&2(1)& \\[-2pt]
      &21&10&3(2)\tablefootmark{b}&3&4(4)&10&6(1)&4&2(1)&--0&3(2)& \\[-2pt]
      &22&10&6(1)\tablefootmark{b}&1&8(1)&10&4(2)&2&3(4)&0&2(2)& \\[-2pt]
      &23&9&6(1)\tablefootmark{b}&2&7(1)&9&8(1)&3&4(1)&--0&2(1)& \\
\noalign{\smallskip}
Cas A &24&--46&80(5)\tablefootmark{b} & 2&1(1) &--47&4(1)&2&9(2) &0&60(11) & Mookerjea \etal\ (\cite{Mookerjea2006b})\\[-2pt]
      &25&--1&10(6)\tablefootmark{b}&1&8(1)&--1&7(1)&3&1(4)&0&60(12)& \\
\noalign{\smallskip}
\hline
\end{tabular}
$$
\tablefoot{\tablefoottext{a}{$J=2{-}1$;} \tablefoottext{b}{$J=1{-}0$}.} 
\end{table}